Massachusetts Institute of Technology Handout 2 6.046J/18.410J: Introduction to Algorithms February 5, 2002 Professors Michel Goemans and Piotr Indyk

Problem Set 0

This problem set is due by 8:00 a.m. on Wednesday, February 6.

Problem 0-1. Registration Fill out sign-up sheet on the course web page at

http://theory.lcs.mit.edu/classes/6.046

before 8:00 a.m. on Wednesday, February 6.

Signing up is a requirement of the course. Your homework and exams will be ignored if you have not signed up. The information you provide will help the course staff to get to know you better, assign you to a recitation, and create a mailing list and a course directory. We will send out confirmation on Wednesday evening by e-mail. If you do not receive e-mail from us by Thursday morning, send e-mail to greenie@mit.edu. If, for any reason, you must sign up late, please see a TA.